Author's response to reviews

Title: Fluid Lavage in Patients with Open Fracture Wounds (FLOW) An International Survey of 984 Surgeons

Authors:

Brad Petrisor (petrisor@Hhsc.ca)
Kyle Jeray (kjeray@usc.com)
Emil Schemitsch (schemitsch@thewire.com)
Beate Hanson (beate.hanson@aofoundation.org)
Sheila Sprague (sprags@mcmaster.ca)
David Sanders (david.sanders@uwo.org)
Mohit Bhandari (bhandam@mcmaster.ca)

Version: 3 Date: 31 December 2007

Author's response to reviews: see over
RESPONSE TO REVIEWS

REVIEWER COMMENT

On page 9, lines 313 and 314: The reference on line 313 should be only to reference no. 33 (Skedros et al., 2006), and the reference on line 314 should be only to reference no. 32 (Skedros et al., 2007). Note also that there should be three authors in reference no. 32; the authors are J.G. Skedros, K.J. Hunt, and T.C. Pitts.

Response: We have corrected as suggested.